Author's response to reviews

Title: Development and evaluation of an intervention aiming to reduce fatigue in airline pilots. Design of a randomised controlled trial.

Authors:

Alwin van Drongelen (a.vandrongelen@vumc.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)
Hynek Hlobil (hhlobil.health@klm.com)
Tjabe Smid (tjabe.smid@klm.com)
Cécile RL Boot (crl.boot@vumc.nl)

Version: 2 Date: 13 September 2012

Author's response to reviews: see over
September 13th, 2012

Dear editor,

After receiving your email (03-09-2012) regarding our manuscript “Development and evaluation of an intervention aiming to reduce fatigue in airline pilots. Design of a randomised controlled trial.” we made the following changes to the manuscript:

(1) We included the Trial Registration Number at the end of the Abstract in the submission system.
(2) We added a sentence whether a written informed consent for participation in the study was obtained from the participants:

“By filling in the questionnaire the pilots agree to participate in the study, no additional written informed consent will be obtained. After the researchers receive the filled in questionnaire, participants will be allocated to either the control or intervention group.”

Yours sincerely,

Drs. A. van Drongelen
Department of Public and Occupational Health
VU University Medical Center
Van der Boechorststraat 7
NL-1081 BT Amsterdam
T: +31 20 4448336
F: +31 20 4448387
Email: a.vandrongelen@vumc.nl